Author’s response to reviews

Title: Are physiotherapists employing person-centred care for people with dementia? An exploratory qualitative study examining the experiences of people with dementia and their carers

Authors:
Abigail Hall (ah577@exeter.ac.uk)
Lisa Burrows (lisa.burrows@plymouth.ac.uk)
Iain Lang (i.lang@exeter.ac.uk)
Ruth Endacott (ruth.endacott@plymouth.ac.uk)
Vicki Goodwin (v.goodwin@exeter.ac.uk)

Version: 3 Date: 23 Feb 2018

Author’s response to reviews:

Dear Editor,

Thank you for your email. All the revisions you have requested have been made and now reflected in the manuscript.

All participants gave written, informed consent. This was a pre-requisite of taking part in the study (line 129). LB’s contribution to the paper has been corrected.

The requested files have been removed from the submission and the Table and Figure have been uploaded separately. I hope this meets the requirements.

Kind regards,

Abi